Author’s response to reviews

Title: Effect of interscalene block on intraocular pressure and ocular perfusion pressure

Authors:

Betul Basaran (betulbasaran1@yahoo.com)

Aysun Ankay Yılbas (aysunankay@hotmail.com)

Zeki Gultekin (mzegultekin@yahoo.com)

Version: 3 Date: 12 Oct 2017

Author’s response to editor

Title: Effect of interscalene block on intraocular pressure and ocular perfusion pressure

Authors: Betul Basaran (betulbasaran1@yahoo.com) Aysun Ankay Yılbas (aysunankay@hotmail.com) Zeki Gultekin (mzegultekin@yahoo.com)

Dear Editor,

We appreciate the editor and reviewers for their comments on our manuscript. We have made related revisions according to your comments. We look forward to hearing from you.

Best Regards,

Betul Basaran, MD

1. All authors have signed authorship change form and the form has been sent.
2. E-mail addresses of all authors have been added to title page.
3. List of abbreviations has been included to our manuscript.
4. Figure 3 is created by an artist. It is under copyright that include permissions about its adaptation and usage.
5. Table names were corrected according to your comments.
6. We have corrected grammatical errors.